Author’s response to reviews

Title: Discrepancy between radiological and pathological size of renal masses

Authors:

Nicola N Jeffery (nikki.jeffery@gmail.com)
Norbert Douek (norbert_do@hotmail.com)
Ding Y Guo (drdingguo@hotmail.com)
Manish I Patel (mpatel@med.usyd.edu.au)

Version: 3 Date: 6 February 2011

Author’s response to reviews:

Dear editor and reviewers,

We are delighted that our manuscript is accepted for publication in BMC Urology.

We have made the following changes to our manuscript, as requested:
- Page 14: We have stated that all authors have read and approved the final manuscript.
- Page 21: We have added a title for Figure 1: "Scatter plot of pathological tumor size against radiological tumor size"
- We have checked our references, and updated references where appropriate (i.e. reference 26).

- We have removed all tracking and highlighting from the manuscript.

Please let us know if there are any further changes that need to be made.

Regards,
Dr Nicola Jeffery
Dr Norbert Douek
Dr Ding Guo
A/Prof Manish Patel